Math 132 Winter 2004 R. Schultz

UPDATED GENERAL INFORMATION — JANUARY 26, 2004

Since it is unlikely that the second homework assignment can be returned before the first midterm next Monday, I have posted an enlarged set of solutions which includes the answers to the assigned exercises from Fraleigh and Beauregard. These are in the document

solutions2.pdf.

Here is the third homework assignment, which is due in class on Monday, February 2 2004. “F&B” refers to the course text by Fraleigh and Beauregard.

Exercises 2, 6, 14a and 34 beginning on p. 336 of F&B.

Exercise 2 beginning on p. 347 of F&B.

Exercise 26 beginning on p. 368 of F&B.

Additional Exercises 3ab and 4b for Section II.2

The material in this assignment is part of the material for the first midterm examination, which will take place on Monday, February 2. Since the homework is only due on the day of the examination, I will post a document with references to answers for similar exercises from the text as well as hints for some of the additional exercises. This document will also contain additional suggestions for review.